

Contents lists available at ScienceDirect

Digital Business



journal homepage: www.elsevier.com/locate/digbus

Success factors of global virtual teamwork: A social capital perspective

Tim Heubeck^{*}, Anne-Sophie Storz, Reinhard Meckl

Faculty of Law, Business, and Economics, Chair of International Management, University of Bayreuth, Universitätsstraße 30, 95447 Bayreuth, Germany

ARTICLE INFO

Keywords: Social capital Global virtual teamwork Team effectiveness Information and communication technologies

ABSTRACT

Global virtual teamwork has emerged as a cornerstone of collaborative teamwork in today's work landscape, characterized by cross-border collaboration using digital technologies. Although research has started to investigate the mechanisms underlying effective teamwork through information and communication technologies (ICTs), an often-neglected dimension is the pivotal social context within which this collaboration unfolds. To address this research gap, this study adopts a social capital lens on teamwork in proposing social capital as a multidimensional mediator between the usage of ICTs and team effectiveness. The research model was tested using questionnaire data from 271 technologically sensitive and aware companies in the DACH region (Germany, Austria, Switzerland). The findings reveal that using multiple ICTs alone does not suffice for effective global virtual teamwork. Therefore, this study highlights the significance of social capital for the effectiveness of global virtual teamwork, which holds significant implications for theory and practice.

1. Introduction

Advances in information and communication technologies (ICTs), coupled with the increasing prevalence of globalization, have driven organizations to rely on digital technologies for globally distributed collaboration—so-called global virtual teamwork (Tavoletti & Taras, 2023). Due to several benefits, such as time and cost savings, talent pooling, and increased team diversity, global virtual teams were already well established in today's workplace before the COVID-19 pandemic (Jimenez, Boehe, Taras, & Caprar, 2017; Kozlowski & Ilgen, 2006; Tavoletti & Taras, 2023). However, global virtual teamwork also presents significant challenges that can threaten team effectiveness. Reliance on ICTs reduces media richness and delays feedback, including an increased risk of miscommunication and misunderstanding (Bilotta et al., 2021; Klitmøller & Lauring, 2013; Mangla, 2021).

Based on the premise that a single medium cannot support the two fundamental communication processes of conveyance (i.e., sharing new information) and convergence (i.e., reaching a common understanding), we introduce the concept of *ICT usage*, which refers to the frequent and strategic use of multiple ICTs (Dennis, Fuller, & Valacich, 2008; Niinimäki, Piri, Lassenius, & Paasivaara, 2012). We propose that high levels of ICT usage will increase team effectiveness, which consists of performance, longevity, and satisfaction (Dennis et al., 2008; Hackman, 1987).

Nevertheless, rather than focusing solely on the characteristics of

ICT, it is essential to consider the embeddedness of communication in a social context. Considering the integration of the individual team member into a collective unit within a larger organizational context, we approach global virtual teamwork from a social capital perspective (Kozlowski & Bell, 2013; Kozlowski & Ilgen, 2006). Social capital refers to networks of relationships and the assets embedded within them. It offers several advantages, including increased resource and information exchange, trust, reciprocity, and collaboration (Adler & Kwon, 2002; Adner & Helfat, 2003; Heubeck & Meckl, 2022). Based on the benefits of social capital, we suggest that the level of ICT usage influences team effectiveness as it allows for the use of the intermediate advantages of social capital (Cao, Guo, Vogel, & Zhang, 2016; Staber, 2006). This study is a pioneering effort to examine the role of social capital as a multidimensional mediator between ICT usage and the effectiveness of global virtual teamwork. By integrating social capital into the study of teamwork, we shed light on the importance of social relationships, networks, and assets in global virtual teamwork settings.

Furthermore, previous studies, such as Ahuja (2000), have often failed to consider the multidimensional nature of social capital, disregarding the unique implications of its various dimensions. To investigate the relationship between ICT usage and the effectiveness of global virtual teamwork, we analyze the impact of social capital's structural, cognitive, and relational dimensions. Our research question is, therefore, the following:

* Corresponding author. *E-mail address*: tim.heubeck@uni-bayreuth.de (T. Heubeck).

https://doi.org/10.1016/j.digbus.2024.100081

Received 5 December 2023; Received in revised form 2 June 2024; Accepted 4 June 2024 Available online 7 June 2024 2666-9544/© 2024 The Authors Published by Elsevier B V. This is an open access article under the CC

2666-9544/© 2024 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

RQ: How do the three dimensions of social capital (structural, cognitive, and relational) influence the relationship between ICT usage and the effectiveness of global virtual teamwork?

To explore the complex relationships between ICT usage, social capital, and team effectiveness in global virtual teamwork settings, we test our research hypotheses on a sample of technologically sensitive and aware companies in the DACH region (Germany, Austria, and Switzerland).

Our study's findings indicate that ICT usage does not necessarily enhance team effectiveness in a global virtual teamwork setting, which stems from a compensatory effect between frequency and variety of ICT usage (Badir, Büchel, & Tucci, 2012; Straube, Meinecke, Schneider, & Kauffeld, 2018; Walther & Parks, 2002). By considering the social context of ICT usage in global virtual teamwork, we demonstrate that structural social capital represents the mechanism through which ICT usage can facilitate the effectiveness of global virtual teamwork (Huysman & Wulf, 2006; Riemer & Klein, 2008). Thus, our findings contribute to the multidimensional discourse on social capital, highlighting the critical role of structural social capital in the complex social web surrounding global virtual teamwork. In this sense, we shift the focus from intercultural team composition (Earley & Mosakowski, 2000; Straube et al., 2018) as well as from a technologically deterministic perspective that focuses on the number of ICTs as determinants of team effectiveness (Boland & Tenkasi, 1995; Huysman & Wulf, 2006) to a perspective that emphasizes the critical role of socialization mechanism in global virtual teamwork (Pianese, Errichiello, & Da Cunha, 2023).

The unique status of structural social capital is also reflected in the absence of intercorrelations with other dimensions of social capital. This finding could also be attributed to the limited ability of ICT usage and intercultural team composition (Hofstede, 2005) to build cognitive and relational social capital (Tsai & Ghoshal, 1998). Nevertheless, while we demonstrate that the positive effect of ICT usage on team effectiveness is driven by structural social capital, our findings highlight that the three dimensions collectively lead to the most beneficial outcomes of ICT usage on team effectiveness in virtual global teamwork. Through these findings, our study significantly advances the theoretical discourse on ICT usage as an antecedent to the effectiveness of global virtual teamwork and highlights the critical role of social capital in facilitating team effectiveness.

The remainder of this article is organized into six sections. Section 2 presents the theoretical background, while Section 3 derives the research model and formulates the hypotheses. Section 4 outlines the research methodology, and Section 5 presents the empirical results. Section 6 discusses their theoretical and practical implications and outlines research limitations and recommendations. Section 7 concludes.

2. Theoretical background

2.1. Global virtual teamwork

Teams consist of "(a) two or more individuals who (b) socially interact (face-to-face or, increasingly, virtually); (c) possess one or more common goals; (d) are brought together to perform organizationally relevant tasks; (e) exhibit interdependencies with respect to workflow, goals, and outcomes; (f) have different roles and responsibilities; and (g) are together embedded in an encompassing organizational system, with boundaries and linkages to the broader system context and task environment" (Kozlowski & Ilgen, 2006, p. 79). Global virtual teamwork is a form of team organization that has evolved due to globalization and is made possible by advances in ICTs (Tavoletti & Taras, 2023). The proliferation of ICTs in all workplace areas enables virtual teamwork. Composed of members connected through ICTs and distributed across space and time, virtual teams can have different degrees of virtuality, ranging from face-to-face interaction to full ICT-mediated collaboration (Kozlowski & Ilgen, 2006; Rudolph et al., 2021). The global connotation of virtual teamwork implies that the team is culturally diverse and geographically dispersed (Jarvenpaa & Leidner, 1999). Various synonyms exist for global virtual teamwork, such as multinational and multicultural distributed teams, transnational teams, or multinational working groups (Jimenez et al., 2017). In essence, all of these terms share the definition of Jarvenpaa and Leidner (1999) as "...temporary, culturally diverse, geographically distributed, and electronically communicating working groups" (p. 792).

2.2. ICT usage

Spatial distance and technologically mediated communication are fundamental characteristics of global virtual teamwork (Bell & Kozlowski, 2002). To overcome geographical dispersion, global virtual teamwork employs various ICT tools such as telephone, video conferencing, email, or instant messaging (Niinimäki et al., 2012). This notion is consistent with the assumption of media synchronicity theory that "it is usually best to use several media either simultaneously [...] or in succession" (Dennis et al., 2008, p. 576).

Media synchronicity theory builds on the widely used media richness theory, which refers to media richness-the ability of a medium to convey information effectively and bring about understanding within a given time frame, depending on factors such as immediate feedback, cues and channels, personalization, and language variety (Daft & Lengel, 1986). In contrast, media synchronicity theory refers to "the extent to which a communication environment encourages individuals to work together on the same activity, with the same information, at the same time; i.e., to have a shared focus" (Dennis, Valacich, Speier, & Morris, 1998, p. 48). Media synchronicity theory identifies five media capabilities that determine the level of synchronicity (Dennis et al., 2008). Effective communication requires a match between these media capabilities and the two fundamental communication processes: conveyance (exchanging new information) and *convergence* (a shared understanding) (Dennis et al., 2008; Niinimäki et al., 2012). Unlike media richness theory, media synchronicity theory does not consider particular media more suited for specific tasks. Instead, it emphasizes evaluating tasks in greater detail, considering conveyance and convergence requirements (Parlamis & Dibble, 2019).

According to media synchronicity theory, using media with lower synchronicity should result in better communication performance for conveyance processes, while using media that support higher synchronicity can improve communication performance for convergence processes. Because successful completion of most tasks involving both conveyance and convergence processes, relying on a single medium is inadequate for optimal communication performance. Instead, media synchronicity theory advocates using multiple media simultaneously or in succession (Dennis et al., 2008). We introduce the term ICT usage based on media synchronicity theory's assumption that most tasks involve conveyance and convergence processes. A high level of ICT usage can facilitate team effectiveness, encompassing performance, satisfaction, and longevity. What differentiates the concept of ICT usage from the aforementioned theories is its simultaneous consideration of two dimensions: the frequency and variety of ICT usage. Specifically, ICT usage encompasses commonly used ICTs such as telephone, video conferencing, emails, and instant messaging (Niinimäki et al., 2012). Thus, ICT usage includes technologies such as videoconferencing that are considered rich or synchronous according to the principles of media richness theory or media synchrony theory, as well as less rich or asynchronous ICTs such as emails (Kirkman, Cordery, Mathieu, Rosen, & Kukenberger, 2013).

2.3. Social capital

Social capital benefits individuals, teams, and organizations by providing access to valuable resources within their networks (Coleman, 1988; Staber, 2006). Although various theoretical perspectives on social

capital have emerged, they share two central commonalities. First, social capital is commonly viewed as a resource (Bourdieu, 1986; Lin, 2001; Nahapiet & Ghoshal, 1998; Putnam, Leonardi, & Nanetti, 1994). Second, unlike other forms of capital, such as economic capital, social capital arises from social relations and shared belief systems (Bourdieu, 1986; Coleman, 1988; Lin, 2001; Nahapiet & Ghoshal, 1998).

Unlike previous measurement models, social capital is no longer considered a one-dimensional factor but a multidimensional approach (Fornoni, Arribas, & Vila, 2011). Three dimensions of social capital that shape individual and collective action dominate the literature (Heubeck, 2023). Nahapiet and Ghoshal (1998) distinguish social capital into structural, relational, and cognitive dimensions. First, structural social capital refers to the overall network connection, including who is connected and how they are connected. Central elements of structural social capital include network ties, the arrangement of the network (e.g., density, connectivity, or hierarchy), and the capacity to leverage social capital (i.e., the transferability of social capital to different contexts). Second, cognitive social capital relates to shared interpretations, representations, and systems of meaning, shaping the absorption and processing of information and the creation of knowledge (Nahapiet & Ghoshal, 1998). It becomes evident through shared languages, codes, and narratives that reflect the retention of past experiences and the construction of new meanings (Staber, 2006). Third, relational social capital pertains to the characteristics and content of relationships, encompassing intangible assets developed and utilized through these connections. Facets of relational social capital include trust, norms, obligations, expectations, identity, and identification (Nahapiet & Ghoshal, 1998). Although the three dimensions of social capital are analyzed separately, they are highly interconnected (Nahapiet & Ghoshal, 1998).

3. Hypotheses development

We propose a research model at the team level. As illustrated in Fig. 1, we focus on the mediation effect of social capital between ICT usage and team effectiveness. Effective ICT usage encompasses the frequent utilization of multiple ICTs (Dennis et al., 2008). By adopting a social capital perspective, we argue that ICT usage influences team effectiveness through the intermediate benefits of social capital (Cao et al., 2016; Staber, 2006). Considering the social and relational complexities inherent in global virtual teamwork and the prevailing social landscape, social capital offers a valuable understanding of the functioning of social networks, including teams (Cao et al., 2016; Coleman, 1988).

We adopt Hackman (1987) conceptualization, designed explicitly for entrepreneurial work groups, to measure team effectiveness. It evaluates team effectiveness based on three criteria. *Performance* refers to whether the team's output meets or surpasses the performance standards of recipients or evaluators (Hackman, 1987). *Longevity* assesses whether the social processes employed by teams during their work enhance their members' ability for future collaboration (Wageman, Hackman, & Lehman, 2005). *Satisfaction* evaluates whether group experiences fulfill personal needs rather than frustrate them. This inclusive definition of team effectiveness incorporates social and personal criteria, deviating from traditional approaches that rely solely on objective performance indicators. Objective criteria were deliberately avoided due to the limited applicability of reliable and valid measures and the greater reliance of future teams on evaluations of group performance rather than objective standards (Hackman, 1987).

Effective teamwork often involves concurrently utilizing multiple communication media (Watson-Manheim & Bélanger, 2007). According to Dennis et al. (2008), employing multiple ICTs simultaneously or sequentially is more effective than relying on a single communication medium. This is because a single medium may not adequately support the processes of task convergence and conveyance (Dennis et al., 2008; Niinimäki et al., 2012). Since most tasks and activities involve convergence and conveyance processes, utilizing multiple communication media will likely enhance performance (Parlamis & Dibble, 2019). Furthermore, using only one communication medium can undermine group cohesion (Knight, Pearson, & Hunsinger, 2008), potentially reducing satisfaction and loyalty. Based on these arguments, we propose the following hypothesis:

H1: ICT usage has a positive effect on team effectiveness in global virtual teamwork.

Communication, encompassing ICT usage, is inherently embedded within a social context characterized by interactions and interpersonal relationships (Cao et al., 2016). Thus, we posit that ICT usage is vital in global virtual teamwork by shaping social capital. Social capital can be viewed as an outcome of daily interactions, with frequent interactions fostering the development of structural social capital (Cao, Vogel, Guo, Liu, & Gu, 2012). Additionally, regular communication nurtures confidence in social relationships, facilitating the sharing of mutual understanding and trust (Badir & O'Connor, 2015; Bstieler, 2006; Lee, Saunders, Panteli, & Wang, 2021; Paulraj, Lado, & Chen, 2008). Based on these arguments, we propose the following hypotheses:

H2: ICT usage has a positive effect on social capital in global virtual teamwork.

H2a: ICT usage has a positive effect on structural social capital in global virtual teamwork.

H2b: ICT usage has a positive effect on cognitive social capital in global virtual teamwork.

H2c: ICT usage has a positive effect on relational social capital in global virtual teamwork.

Previous research suggests that social capital is likely to promote team effectiveness (e.g., Andersson, Forsgren, & Holm, 2002; Castro & Roldán, 2013; Chang & Chuang, 2011; Huang, Chen, Ou, Davison, & Hua, 2017; Nahapiet & Ghoshal, 1998; Park & Luo, 2001; Sun, Fang, Lim, & Straub, 2012; Tsai & Ghoshal, 1998). Network ties, a key measure of structural social capital, not only connect individuals but also serve as channels for valuable resources, including knowledge and information (Castro & Roldán, 2013; Huang et al., 2017; Nahapiet & Ghoshal, 1998; W. Wu, 2008). Dense network ties offer several benefits that contribute to improved performance and satisfaction. Social interactions facilitated by network ties reduce the effort and time required for information exchange (Chang & Chuang, 2011; Huang et al., 2017). Moreover, high network density enhances the social aspects of jobs,



Fig. 1. Research model.

increasing job satisfaction (Flap & Völker, 2001; Yamaguchi, 2013). Cognitive social capital promotes resource exchange simultaneously (Villena, Revilla, & Choi, 2011). This shared cognitive framework fosters synergies and reduces conflicts, ultimately enhancing operational and strategic performance (Jones & Taylor, 2012; Nahapiet & Ghoshal, 1998; Villena et al., 2011). Furthermore, a shared language and common interpretation scheme facilitated by cognitive social capital prevent misunderstandings and reduce cognitive effort, contributing to satisfaction (Huang et al., 2017; Sun et al., 2012). Cognitive social capital also promotes longevity through shared visions and goals (M. Wu, Coleman, Abdul Rahaman, & Edziah, 2020; W. Wu, 2008). Finally, the relational dimension of social capital grants access to resources and facilitates their exchange due to greater trust (Castro & Roldán, 2013; Uzzi, 1996). Trust reduces information asymmetries in information exchange by mitigating opportunistic behavior (Castro & Roldán, 2013; Raghupathi & Benbunan-Fich, 2020). Trust-based relationships foster effective team performance by establishing predictable actions and reinforcing cooperation (Cao et al., 2012; Castro & Roldán, 2013; Liu, 2012; Raghupathi & Benbunan-Fich, 2020; Tsai & Ghoshal, 1998; W. Wu, 2008). Similarly, the effect of trust on satisfaction sets expectations for a satisfactory outcome (Gefen & Straub, 2003; Huang et al., 2017). Trust and reciprocity ensure resource exchange through mutual indebtedness, and individuals in reciprocal relationships tend to be more satisfied due to the expectation of equivalent returns in the future (Huang et al., 2017; McLure Wasko & Faraj, 2005). Based on this reasoning, we propose the following hypotheses:

H3: Social capital has a positive effect on team effectiveness in global virtual teams.

H3a: Structural social capital has a positive effect on team effectiveness in global virtual teams.

H3b: Cognitive social capital has a positive effect on team effectiveness in global virtual teams.

H3c: Relational social capital has a positive effect on team effectiveness in global virtual teams.

Building upon the previous hypotheses, we propose that social capital strengthens the relationship between ICT usage and team effectiveness. Focusing on the characteristics of ICTs is insufficient, as it is essential to consider the embeddedness of communication in a social context. Therefore, organizations face the challenge of cultivating social capital, which can be achieved by facilitating new connections among employees, providing dedicated time and space for relationship building, and establishing explicit norms, culture, language, and a shared purpose that influence the formation of team members' values, goals, and expectations (Ali-Hassan, Nevo, & Wade, 2015; Ghoshal & Moran, 1996; Lesser & Storck, 2001).

Studies have demonstrated that social capital mediates the relationship between ICT usage and performance (Ali-Hassan et al., 2015; Cao et al., 2012; Kamboj, Kumar, & Rahman, 2017). Furthermore, social capital has also been shown to mediate the relationship between ICT usage and satisfaction (Chan, 2015; Fu, Sawang, & Sun, 2019; Pang, 2018). By increasing their involvement through ICT usage in both strong and weak social networks, individuals gained access to valuable resources, opportunities, and supportive communities, ultimately enhancing their psychological state (Chen & Li, 2017; Pang, 2018). The preceding discussion underscores the significance of incorporating social capital for the effectiveness of virtual and global virtual teamwork due to its intrinsic connection to the social environment. In light of this rationale, we put forward the following hypothesis.

H4: Social capital mediates the relationship between ICT usage and team effectiveness in global virtual teams.

H4a: Structural social capital mediates the relationship between ICT usage and team effectiveness in global virtual teams.

H4b: Cognitive social capital mediates the relationship between ICT usage and team effectiveness in global virtual teams.

H4c: Relational social capital mediates the relationship between ICT usage and team effectiveness in global virtual teams.

4. Methodology

To investigate the complex relationships between ICT usage, social capital, and team effectiveness in global virtual teamwork settings, we collected data from Industry 4.0 due to their sensitivity and awareness toward technologies. The term "Industry 4.0" refers to companies that successfully implemented "DT [digital transformation] manufacturing, in which interconnected processes and equipment allow the mass customisation of products and responses to market demands" (Grooss, Presser, & Tambo, 2022, p. 1). We focused on Industry 4.0 companies headquartered in the DACH region (Germany, Austria, and Switzerland), as the region has a significant presence of technologically advanced companies with a proactive approach to digital transformation (Heubeck, 2023; Schneider, 2018). To prevent cultural biases in respondents' perceptions, we restricted the sample to individuals with a cultural background from the Germanic Europe cluster, comprising Germany, Austria, Netherlands, and Switzerland. In addition to linguistic similarities, this cluster shares cultural values such as order, straightforwardness, honesty, and loyalty, which play an essential role in society (V. Gupta & Hanges, 2006). Initially, we compiled a company list using the member lists available on the websites of the three industry associations forming *Platform Industry* 4.0. To accurately represent the Industry 4.0 context, we researched the industry sector codes for these companies. We used Bureau van Dijk's Amadeus database, containing financial and business data from approximately 21 million European public and private companies (Bureau van Dijk, 2023), to generate a comprehensive list of companies operating in the selected industries within the DACH region, along with their contact information.

In September and October 2022, we emailed questionnaire invitations to 49,448 companies using the Qualtrics web-based survey tool. We utilized the key informant approach, with each company represented by someone with experience and knowledge of global virtual teamwork. Non-respondents received reminders after three weeks. The questionnaire consisted of two parts. The first part collected information on respondent, team, and company characteristics, while the second part measured the study constructs using a five-point Likert scale ranging from 1 (strongly disagree) to 5 (strongly agree). Certain items were reverse-coded to omit response bias (Price, 1997). Following the key informant approach, all items were based on the individual perceptions of company representatives in the context of global virtual teamwork. The respondents were requested to respond to the questions based on their experiences related to global virtual teamwork.

To ensure respondent validity, respondents were provided definitions of teamwork and global virtual teamwork to ensure response validity. They were then asked three questions: (1) if they worked with team members from other cultures, (2) if team members were located in different countries, and (3) if their team primarily communicated through ICTs. Respondents answering 'no' to any of these questions were unable to proceed with the remaining questionnaire. This was done to guarantee that the collected data pertains solely to experiences with global virtual teamwork. A total of 1082 company representatives answered the questionnaire, giving a response rate of 2.19%. Of the 1082 questionnaires received, 811 were excluded from the analysis for various reasons: firstly, 362 were irrelevant to the focus of the study on global virtual teamwork; secondly, the remaining questionnaires were either not fully completed, or the company respondents did not have a cultural background from the DACH region. The remaining 271 questionnaires were analyzed to ensure a culturally homogenous sample belonging to the so-called Germanic Europe Cluster, encompassing Germany, Austria, the Netherlands, and Switzerland (V. Gupta & Hanges, 2006). This cultural limitation does not mean that respondents only work with other team members from the Germanic Europe Cluster. The pertinent literature supports the adequacy of this sample size for analysis (Comrey & Lee, 1992; Osborne & Costello, 2019).

The final sample consisted of 271 company representatives, with respondents predominantly being German (87.45%), primarily male

(70.48%), and holding university degrees (75.65%). Our sample is a good representation of the Industry 4.0 population (S. Gupta, Modgil, Gunasekaran, & Bag, 2020). The respondents worked in different types of teams, including management, work, project, and parallel teams (S. G. Cohen & Bailey, 1997). Most respondents (66.79%) had the opportunity to meet face-to-face before starting global virtual teamwork. Table 1 provides an overview of the respondents' demographic characteristics.

Following established guidelines for questionnaire development, translation, and validation (e.g., Harkness, Pennell, & Schoua-Glusberg, 2004; Tsang, Royse, & Terkawi, 2017), we conducted a thorough literature search to identify validated measurement scales for the study variables. As the relevant literature was predominantly in English and the respondents were German-speaking, we decided to translate the validated scales into German using a recommended forward and backward translation process.

Initially, two of the authors (native German speakers) and a professional academic translator (native English speaker) independently translated the items into German (Beaton, Bombardier, Guillemin, & Ferraz, 2007; Guillemin, Bombardier, & Beaton, 1993). Subsequently, these translations were independently back-translated from German to English by all three parties (Beaton et al., 2007; Guillemin et al., 1993; Tsang et al., 2017). The forward and backward translations were then thoroughly reviewed and discussed among all the authors to ensure "semantic, idiomatic, experiential, and conceptual equivalence" (Tsang et al., 2017, 84). Minor discrepancies were resolved, and a consensus was reached on the final translations of the items.

The translated and validated questionnaire was subjected to a pilot test involving two sample respondents from the study's target population. Additionally, the third author, an expert in the field, provided additional feedback. The three experts reviewed the questionnaire and provided minor comments on the formulation of the questions. After incorporating their feedback and making necessary adjustments, all experts agreed on the questions' comprehensibility.

4.1. Variable measurements

Appendix A summarizes the measurement scales for latent constructs in their original English version. *Team effectiveness* was assessed using the scale developed by Wageman et al. (2005). According to their definition of team effectiveness, the respective items can be grouped into three dimensions: performance, satisfaction, and longevity of the team, as defined by Hackman (1987). The *social capital* scale used in this study is from Sun et al. (2012) and has been modified to align with the context of teamwork. *ICT usage* was measured using the scale developed by

Table 1

Demographic characteristics of respondents.

Eisenberg, Glikson, and Lisak (2021), which employs a Likert scale ranging from 1 ("did not use at all") to 5 ("used to a great extent"). The mean response on all four items of ICT usage was calculated to capture the frequency of diverse ICT usage in global virtual teamwork, indicating frequent and diverse communication.

Control variables were incorporated at the firm, team, and respondent levels. First, firm size was the natural logarithm of the number of current employees in the firm. Second, team size captured the number of team members. Third, team age represented the duration of the team's existence, including the period before the respondent joined. Fourth, team familiarity assessed whether team members had the opportunity to get to know each other through face-to-face interaction before engaging in global virtual teamwork (yes = 0; no = 1). Fifth, team type was classified based on S. G. Cohen and Bailey (1997) framework, encompassing four categories: work teams, parallel teams, project teams, and management teams. Sixth, respondent age indicated the current age of the respondent. Seventh, respondent sex was a dummy variable based on the German civil status law, which recognizes more than two sexes (Federal Anti-Discrimination Agency, 2018); our survey also included an option to select "diverse" for intersex people (male = 0; female = 1; no respondents identified as diverse). Eighth, respondent team tenure measured the number of months the respondent had been a member of this team. Ninth, respondent education captured the level of education attained by the respondent (no degree = 0; professional education = 1; applied scientific degree = 2; university degree = 3). Finally, *respondent nationality* was coded as follows: German = 1; Austrian = 2; Swiss = 3.

4.2. Factor analysis, construct validity, and reliability

Factor analysis was conducted using *IBM SPSS Statistics Version 29*. Principal axis factor analysis was employed for each theoretically proposed factor, utilizing varimax rotation and Kaiser normalization. The appropriateness of factor analysis was assessed through the Bartlett test of sphericity, the measure of sample adequacy (MSA) criterion, and the Kaiser-Mayer-Olkin (KMO) criterion (Hair, Black, Babin, & Anderson, 2014). The determination of the number of extracted factors was based on the Kaiser-Guttman (KG) criterion, the scree test (Thompson, 2004), and the latent root criterion, which considers factor loadings >0.40 and eigenvalues of at least 1 (Gower, 1966). Factors consisting of a minimum of three items were retained (Hair et al., 2014). Reliability was assessed using Cronbach's alpha, with values >0.70 indicating acceptable reliability (Hair et al., 2014). Validity was evaluated based on the average variance extracted (AVE) and the Fornell-Larcker ratio (FLR), with an AVE >0.50 and an FLR below 1 indicating satisfactory validity (Fornell

Variable			Frequency	%	Cumulated %
Respondent sex	Male		191	70.48	
	Female		80	29.52	100.00
Respondent education	No degree		8	2.95	
	Professional education		29	10.70	13.65
	Applied scientific degree		29	10.70	24.35
	University degree		205	62.00	100.00
	Of which	Bachelor's degree	29	10.70	
		Master's degree	48	17.71	
		Diploma	105	38.75	
		PhD	23	8.49	
Respondent nationality	German		237	87.45	
	Austrian		21	7.75	95.20
	Swiss		13	4.80	100.00
Team type	Work team		88	32.47	
	Parallel team		34	12.55	45.02
	Project team		59	18.08	63.10
	Management team		100	36.90	100.00
Team familiarity	Yes		181	66.79	
	No		90	33.21	100.00

Notes N = 271.

& Larcker, 1981). Convergent validity could also be established if the AVE fell between 0.40 and 0.50 and Cronbach's alpha coefficients exceeded 0.60 (Fornell & Larcker, 1981).

The factor analysis results are summarized in Appendix B, with detailed explanations provided below. The factor analysis for social capital met all the defined criteria. The three-factor solution aligned with the theoretically proposed dimensions of structural, cognitive, and relational social capital, extracting one factor for each dimension. The measurement scale for relational social capital underwent a slight modification, as Item 2 was removed to enhance the factor's reliability. The factor analysis for team effectiveness also satisfied all the defined criteria. Three factors were initially extracted for the performance dimension, but Item 3, which loaded onto a separate factor, was subsequently eliminated from the scale. Consequently, team performance comprised two factors. Regarding team satisfaction, three factors were retained, although Item 4 was excluded from the scale as it did not load onto any of the extracted factors. Despite the second factor comprising only two items, it was retained for further analysis due to its contribution to content validity and fulfillment of quality criteria (Gosling, Rentfrow, & Swann, 2003). Finally, two factors were extracted for team longevity, and no modifications were made to the measurement scale.

To address common method bias, various measures recommended in the literature were implemented (MacKenzie & Podsakoff, 2012; Podsakoff, MacKenzie, Lee, & Podsakoff, 2003). First, social desirability bias was minimized by ensuring participant confidentiality, clarifying that no additional personal data would be recorded through the online questionnaire, and guaranteeing that the statistical methods employed would not enable individual respondent identification. As previously mentioned, the questionnaire underwent a pilot test to ensure the clarity and comprehensibility of the questions (MacKenzie & Podsakoff, 2012). Different scales and reverse-coded items were also used to mitigate common method bias (MacKenzie & Podsakoff, 2012; Podsakoff et al., 2003). Harman's single-factor test, which loads all items measuring latent variables onto a single common factor, was employed to examine potential common method bias. The single factor explained a total variance of 26.98%, well below the commonly accepted threshold of 50% (Harman, 1976; Podsakoff et al., 2003). Based on the outlined procedure and the assessment using Harman's test, it is unlikely that common method bias significantly influenced the research data in this study.

4.3. Analysis method

The proposed research model posited that social capital acts as a strengthening mediator in the relationship between ICT usage and team effectiveness. The bootstrapping method was employed using Hayes' (2021) PROCESS macro for SPSS to test this model. Confidence intervals were calculated based on 5000 bootstrap samples, and heteroscedasticity-robust inference HC4 (Cribari-Neto) was utilized for standard errors. The robustness of significant mediation effects was further assessed using Sobel's test (Baron & Kenny, 1986; Sobel, 1982). According to Baron and Kenny (1986), mediation is established if (1) the independent variable significantly impacts the mediating variable and (2) the mediating variable significantly affects the dependent variable in the full regression model. It should be noted that a significant total effect of the independent variable on the dependent variable is not always necessary for mediation (Hayes, 2009; Zhao, Lynch, & Chen, 2010).

5. Results

Table 2 presents descriptive statistics and bivariate results, while Appendix C displays regression results with social capital as the independent and mediating variable. Regression results for the three social capital dimensions are presented in Appendix D. The effect sizes and confidence intervals obtained through bootstrapping, as well as Sobel's test results, are compiled in Appendix E. A summary of the hypothesis test results can be found in Table 3.

The presence of multicollinearity was examined, and the maximum variance inflation factor of 1.411 was found to be well below critical values (Johnston, Jones, & Manley, 2018; Menard, 2002). Effect sizes are classified according to J. Cohen (1988) criteria: $\beta > 0.02$ (weak effect), $\beta > 0.15$ (moderate effect), and $\beta > 0.35$ (strong effect). Significance levels are ranked as follows: p < 0.001 (extremely significant), p < 0.05 (significant), and p > 0.05 (insignificant) (Wooldridge, 2019).

Hypothesis 1 posited that ICT usage increases team effectiveness. However, the findings do not support this hypothesis as the direct effect is not significant (H1: b = 0.061, se = 0.046, p = 0.305). Thus, Hypothesis 1 is rejected.

Hypothesis 2 proposed that ICT usage increases social capital. The results support the general benefits of ICT usage for social capital, as it shows a highly significant and moderate effect (H2: b = 0.217, se = 0.071, p = 0.003). The inclusion of ICT usage in the model improves the explanatory power by 3.6%. Thus, Hypothesis 2 is supported.

Hypotheses 2a–c further conjectured that ICT usage positively affects the three social capital dimensions. The results provide support for Hypothesis 2a, indicating an extremely significant and strong positive effect of ICT usage on structural social capital (H2a: b = 0.583, se =0.104, p < 0.001). However, the effect of ICT usage on cognitive social capital is negative and insignificant (H2b: b = -0.019, se = 0.117, p =0.870, and the effect on relational social capital is positive but insignificant (H2c: b = 0.086, se = 0.091, p = 0.345). Therefore, Hypothesis 2a is supported, while Hypotheses 2b and 2c are rejected.

Hypothesis 3 suggested that social capital positively affects team effectiveness, and the results support this hypothesis with a strong positive and extremely significant coefficient (H1: b = 0.707, se = 0.048, p < 0.001). Including the study variables in the regression model significantly increases explanatory power from 0.031 to 0.514.

Relatedly, Hypothesis 3a–c proposed that the three social capital dimensions increase team effectiveness. The results support this hypothesis for each dimension, indicating that structural social capital has an extremely significant and moderate effect (H3a: b = 0.190, se = 0.031, p < 0.001), cognitive social capital has an extremely significant and moderate effect (H3b: b = 0.174, se = 0.029, p < 0.001), and relational social capital has an extremely significant and strong positive effect (H3c: b = 0.343, se = 0.036, p < 0.001) on team effectiveness.

Hypothesis 4 consolidated all previous hypotheses, proposing that social capital acts as a strengthening mediator between ICT usage and team effectiveness. The bootstrapping results confirm a highly significant and moderately positive indirect effect of ICT usage on team effectiveness via social capital (H4: b = 0.153, se = 0.069, 99% *CI*: 0.023, 0.290). Sobel's test also confirms the significance of this indirect effect (b = 0.148, p = 0.004). The total effect of ICT usage on team effectiveness is highly significant and positive (b = 0.214, se = 0.069, p = 0.002), while the direct effect is insignificant (b = 0.061, se = 0.046, p = 0.190). Thus, the results support Hypothesis 4 by confirming the indirect-only mediation effect of ICT usage on team effectiveness via social capital.

Three additional hypotheses were proposed to investigate this indirect effect further, suggesting the relationship between ICT usage and team effectiveness through the three social capital dimensions. The data confirm Hypothesis 4a, showing that structural social capital acts as a strengthening mediator between ICT usage and team effectiveness (H4a: b = 0.111, se = 0.026, 99% *CI*: 0.052, 0.189). Sobel's test provides additional support for Hypothesis 4a (b = 0.153, p = 0.003). However, the data do not support Hypotheses 4b and 4c. The indirect effect of ICT usage on team effectiveness via cognitive social capital is negative yet insignificant (H4b: b = -0.003, se = 0.021, 90% *CI*: -0.037, 0.030). Sobel's test confirms the insignificance of this effect (b = -0.003, p = 0.872). Similarly, the indirect effect of ICT usage on team effectiveness through relational social capital is positive yet insignificant (H4c: b = 0.029, se = 0.031, 90% *CI*: -0.037, 0.030). Sobel's test confirms the

 \checkmark

Fable Descri	2 ptive statistics a	and bivari	ate results	5.															
	Variable	Mean	SD	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
1	Team effectiveness	0.010	0.527	1															
2	Social capital	0.022	0.523	0.702***	1														
3	Structural social capital	0.042	0.848	0.373***	0.608***	1													
4	Cognitive social capital	0.008	0.864	0.359***	0.629***	0.029	1												
5	Relational social capital	0.017	0.804	0.590***	0.634***	0.100	0.122*	1											
6	ICT usage	3.705	0.524	0.194**	0.209***	0.373***	-0.030	0.047	1										
7	Firm size	4.596	1.950	-0.052	-0.115	-0.113	-0.028	-0.075	-0.004	1									
8	Team size	17.625	26.847	0.021	0.026	0.124*	-0.074	-0.001	0.105	0.105	1								
9	Team age	70.033	89.302	0.041	0.051	0.120*	-0.038	0.014	0.168**	0.028	0.225***	1							
10	Team familiarity	0.332	0.472	-0.069	-0.151*	-0.227***	-0.031	-0.023	-0.070	0.148*	-0.019	-0.035	1						
11	Team type	2.594	1.279	-0.079	-0.073	-0.008	-0.054	-0.075	0.126*	0.056	0.191**	0.025	-0.169**	1					
12	Respondent age	48.435	10.426	0.128*	0.079	-0.050	0.079	0.123*	-0.020	-0.076	0.100	0.117	-0.086	0.033	1				
13	Respondent sex	0.295	0.457	-0.022	-0.036	-0.017	0.027	-0.080	-0.099	-0.032	-0.138*	-0.084	0.162**	-0.092	-0.226***	1			
14	Respondent education	2.590	0.797	-0.010	0.034	0.015	0.037	0.010	0.035	0.016	0.055	0.001	0.048	0.091	0.001	-0.104	1		
15	Respondent nationality	1.173	0.490	-0.023	-0.065	-0.084	-0.005	-0.032	0.085	0.061	-0.047	0.011	0.006	0.024	-0.053	-0.031	0.012	1	
16	Respondent team tenure	87.915	99.668	0.042	0.138*	0.114	0.036	0.110	0.089	-0.129*	0.078	0.374***	-0.160**	0.100	0.338***	-0.192**	-0.089	0.072	1

Notes: SD = Standard deviation; ***p < 0.001, **p < 0.01, *p < 0.05; N = 271.

Table 3

Summary of hypotheses test results.

Hypot	thesis	Result
H1	ICT usage has a positive effect on team effectiveness in global virtual teamwork.	Rejected
H2	ICT usage has a positive effect on social capital in global virtual teamwork.	Supported
H2a	ICT usage has a positive effect on structural social capital in global virtual teamwork.	Supported
H2b	ICT usage has a positive effect on cognitive social capital in global virtual teamwork.	Rejected
H2c	ICT usage has a positive effect on relational social capital in global virtual teamwork.	Rejected
H3	Social capital has a positive effect on team effectiveness in global virtual teams.	Supported
НЗа	Structural social capital has a positive effect on team effectiveness in global virtual teams.	Supported
НЗb	Cognitive social capital has a positive effect on team effectiveness in global virtual teams.	Supported
НЗс	Relational social capital has a positive effect on team effectiveness in global virtual teams.	Supported
H4	Social capital mediates the relationship between ICT usage and team effectiveness in global virtual teams.	Supported
H4a	Structural social capital mediates the relationship between ICT usage and team effectiveness in global virtual teams.	Supported
H4b	Cognitive social capital mediates the relationship between ICT usage and team effectiveness in global virtual teams.	Rejected
H4c	Relational social capital mediates the relationship between ICT usage and team effectiveness in global virtual teams.	Rejected

insignificance of this effect (b = 0.029, p = 0.349). A summary of the hypothesis test results can be found in Table 3.

6. Discussion and contributions

In the current age of digitalization and globalization, the evolving work landscape necessitates understanding how successful global virtual teamwork can be achieved with the help of ICTs. Due to the embeddedness of teams in a social context, where social interactions and interpersonal relationships are essential, it is not sufficient to study only the use of ICTs. Research requires a profound comprehension of how recognizing social context can leverage ICTs to increase the effectiveness of global virtual teamwork.

To fill this research gap, we employed social capital, which distinguishes between the structural, cognitive, and relational dimensions, as multidimensional mediators in the relationship between ICT usage and team effectiveness, encompassing performance, satisfaction, and longevity (Hackman, 1987; Nahapiet & Ghoshal, 1998).

6.1. Theoretical contributions

The empirical findings enhance our understanding of the mediating effects of individual social capital dimensions between adopting ICTs and team effectiveness. From these results, we derive four main theoretical contributions and managerial implications, which are summarized in Table 4.

First, our findings are consistent with previous research (Parlamis & Dibble, 2019; Straube et al., 2018) by showing that in global virtual teamwork, ICT usage advice to use multiple ICTs frequently does not increase team effectiveness. One potential explanation for this finding is the compensatory effect between media richness and communication intensity, as defined by Straube et al. (2018) as "a subjective perception of the informational value that is conveyed through communication" (p. 689). Compensatory adaptation theory suggests that ICTs interfere with the successful transmission of messages (Kock, 1998; Straube et al., 2018). Rather than relying on richer ICTs, these limitations can be overcome through intensive communication activities related to information richness. For example, when using email, it is possible to explain a complex problem in more detail. Since intensive communication using rich ICTs can be perceived as burdensome by stakeholders, the choice of richer ICTs is not inherently better, as efficient communication is characterized by the lowest possible costs (Badir et al., 2012; Straube et al., 2018; Walther & Parks, 2002). Accordingly, combining high media richness and low communication intensity or high communication intensity and low media richness is considered the most advantageous. Applied to the concept of ICT usage, this implies that, contrary to initial assumptions, the frequent use of different ICTs does not necessarily lead to increased team effectiveness. Conversely, some participants may perceive it as a burden. Rather than focusing on the variety of ICTs utilized, directing attention toward the richness of the individual medium is more beneficial. In the previous conception of ICT usage, the individual ICT were considered to be of equal value. Nevertheless, the compensatory adaptation theory explains that individual ICTs are differentially suited to increasing the effectiveness of global virtual teams, depending on the frequency of their use. According to the compensatory adaptation theory, a balance should be struck between the frequency and media richness of ICT. In this case, this implies that for richer media (e.g., video conferencing), global virtual team members communicate less frequently. Conversely, for less rich media (e.g., email), it is important to communicate more frequently.

The insignificant result challenges the initial assumptions regarding

Table 4

Summary of main findings and their theoretical contributions and managerial implications.

Relationship	Finding	Theoretical contribution	Managerial implications
ICT usage → team effectiveness	High ICT usage does not increase team effectiveness in global virtual teams.	There is a compensating effect between the frequency and media richness of ICTs used.	 Global virtual teams can enhance team effectiveness by: Communicating less frequently with richer ICT (e.g., video-conferencing) or Communicating more frequently with less rich ICT (e.g., e-mail).
ICT usage → social capital	ICT usage only increases structural social capital in global virtual teams; building cognitive and relational social capital is only possible to a limited extent in global virtual teams	There are several reasons why the construction of cognitive and relational social capital in global virtual teams is impeded. These include: • Different cultural backgrounds of team members, • A technologically deterministic perspective on ICT usage, and • A lack of focus on planning and innovating organizational structures, management practices, and resources that promote sharing norms, beliefs, and values.	 Global virtual teams can build cognitive and relational social capital through organizational redesign. This can be achieved by: Utilizing brokers to connect individuals and By providing time and a shared physical space for global virtual teamwork.
Social capital → team effectiveness	Social capital increases team effectiveness in global virtual teamwork.	Team effectiveness is enhanced when all dimensions of social capital are considered, emphasizing the direct impact of each dimension.	

the relationship between ICT usage and team effectiveness. It underscores the importance of considering the social context, which shapes the ability of communication to increase the effectiveness of global virtual teams. This finding aligns with this study's social capital perspective. Besides social capital, team composition also shapes social context regarding demographic characteristics (Straube et al., 2018). In the case of global virtual teams, as in the current study, this also includes their interculturality. As already mentioned, social context determines the type of communication. Building on the findings of Straube et al. (2018), it can be concluded that such teams suffer from a high expression of ICT usage but instead benefit from a balanced communication mix between the frequency and media richness of ICTs used. Consequently, the social context cannot be disregarded when assessing the effectiveness of ICTs.

Second, the partial mediation of the relationship between ICT usage and team effectiveness suggests that the contribution of ICT to social capital formation is limited (e.g., D. Cohen & Prusak, 2001; Riemer & Klein, 2008). Consistent with prior research, most ICT applications are primarily suited for building structural social capital (Huysman & Wulf, 2006; Riemer & Klein, 2008). These applications provide a technological infrastructure that enables human actors to connect, communicate, and collaborate, thereby offering the necessary infrastructure and creating opportunities conducive to knowledge exchange (Huysman & Wulf, 2006). On the contrary, ICT effectiveness fosters cognitive and relational social capital to a limited extent (Huysman & Wulf, 2006; Riemer & Klein, 2008). Research examining ICT's role in enhancing cognitive and relational social capital is scarce (Huysman & Wulf, 2006).

There are several reasons for the insignificance of ICT usage-cognitive social capital and ICT usage-relational social capital relationships. One potential explanation for this phenomenon is the intercultural composition of global virtual teams manifested in diverse values and norms (Hofstede, 2005). The study participants confirmed that they collaborate with individuals from diverse cultures across various countries, primarily through ICTs. The presence of individuals from different cultural backgrounds, or more generally diverse backgrounds, impairs the development of cognitive social capital regarding shared understanding (Earley & Mosakowski, 2000; Straube et al., 2018). This, and the tendency to trust within the team, subsequently affects the team's performance and effectiveness (Earley & Mosakowski, 2000). However, the current research model does not account for the degree to which interculturality is emphasized. For future research, it would be beneficial to explicitly incorporate the degree of interculturality.

Another reason for the insignificance of ICT usage-cognitive social capital and ICT usage-relational social capital relationships is the technologically deterministic view of ICTs. Huysman and Wulf (2006) argue that most ICTs focus neither on interpersonal relationships nor reducing the cognitive distance between the actors. A socio-technical perspective may be more appropriate for assessing how social capital can design ICTs than considering social capital as a result of ICT usage. This perspective might provide significant results because, according to the socio-technical perspective, ICTs can only be used for knowledge sharing if the design of ICTs preserves the integrity of the community in which the knowledge is embedded (Boland & Tenkasi, 1995; Huysman & Wulf, 2006). This study's participants corroborated that they primarily collaborate through the use of ICTs. However, the current research model does not consider how these ICTs facilitate the formation of relational and cognitive capital. It would be beneficial to conduct further research with a more precise query on the properties of the ICT used.

Finally, it can be argued that the narrow focus on ICT usage may be responsible for insufficient cognitive and relational social capital development. Companies should not only focus on the amount of ICT used but also on planning and innovating organizational structures, management practices, and resources to support the transition of companies in the post-COVID-19 pandemic era. According to Pianese et al. (2023), forming trusting relationships—a component of relational social capital—is also possible in a virtual context. However, to achieve this, organizations must implement effective socialization mechanisms that promote sharing norms, beliefs, and values among employees. The function of leaders is to encourage innovative processes and cultivate trustworthy relationships among individuals who may not have met in person (Pfister, 2009; Pianese et al., 2023). These measures are essential for coherence and consistency in the attitudes and behaviors of team members to create a shared "culture of control" (Pianese et al., 2023, p. 337).

Third, this study provides empirical support for the positive impact of increased social capital on team effectiveness, which is consistent with the findings from previous studies (e.g., Andersson et al., 2002; Castro & Roldán, 2013; Chang & Chuang, 2011; Huang et al., 2017; Nahapiet & Ghoshal, 1998; Park & Luo, 2001; Sun et al., 2012; Tsai & Ghoshal, 1998). Thus, our study challenges the research stream that argues for a negative (e.g., Rowley, Behrens, & Krackhardt, 2000) or a curvilinear relationship (e.g., Villena et al., 2011) between the expression of social capital and performance. Concerns that high or excessive expressions of social capital may lead to adverse outcomes such as loss of objectivity, opportunism, ineffective decision-making, and costly investments (Gargiulo & Benassi, 1999; Villena et al., 2011) can be refuted in the context of this study.

Therefore, this study finds support for the positive effect of social capital on team effectiveness in a virtual and global context, highlighting its effectiveness in distributed collaboration and the management of complex tasks (Badrinarayanan, Madhavaram, & Granot, 2011; Riemer & Klein, 2008). This study demonstrates that team effectiveness is enhanced when all dimensions of social capital are considered, supporting previous research that emphasizes the direct impact of specific dimensions (e.g., Jarvenpaa, Shaw, & Staples, 2004). Contrary to prior research (e.g., Gedajlovic, Honig, Moore, Payne, & Wright, 2013; Sun et al., 2012), it is evident that the individual dimensions do not operate in a specific sequential order. Unlike previous research, this study considered all dimensions of social capital and Hackman (1987) three criteria for team effectiveness: performance, satisfaction, and longevity. The specificity of this study lies in the research context of global virtual teamwork. Although forming social capital in the virtual environment is sometimes difficult, a high level of social capital and its dimensions positively affect team effectiveness. Similarly, regarding cultural differences, social capital and its components directly affect team effectiveness.

6.2. Managerial implications

The findings of this study hold significant implications for managers, which are summarized in Table 4.

In the first instance, managers must be mindful that the advice to utilize various ICTs frequently is not necessarily applicable in global virtual teams, where the social context (e.g., intercultural considerations) must be considered. It is not always beneficial for every team member to be in frequent contact via different ICTs with every colleague, particularly when such contact may be perceived as excessive. Following the principles of compensatory adaptation theory, global virtual teams should communicate less frequently through richer media (e.g., video conferencing), while conversely, less rich media should be employed with more frequency. Clear and concise communication, with established rules governing ICT use, can facilitate effective global virtual teamwork. Consequently, every team can enhance its performance with a balanced mix of frequency and media richness of ICTs (Straube et al., 2018).

Second, frequent use of different ICTs leads to the development of structural social capital in the form of strong relationships, frequent interactions, familiarity on a personal level, and much time spent together (Nahapiet & Ghoshal, 1998; Sun et al., 2012).

Concerning the relationships between ICT usage and cognitive social

capital, as well as ICT usage and relational social capital, it is evident that organizational redesign is necessary. One possible solution is using brokers, whose main task is connecting individuals. Knowledge brokers within the organization can facilitate the development of cognitive social capital. Their primary role is to act as intermediaries between various social communities rather than connecting the social and technical systems (Huysman & Wulf, 2006). The practice of knowledgesharing research has identified several different types of such "sociotechnical brokers." For instance, some serve as "reviewers" and assess the quality of the knowledge base. There are also "boosters" who facilitate connections between individuals. "Commuters" navigate between users and content providers at the front and back offices. Finally, "experts" seek additional knowledge beyond what is available in the repository (Huysman & Wulf, 2006).

Regarding relational social capital, "trust brokers" (Pianese et al., 2023, p. 337) can help mediate trust between unconnected individuals. Alternatively, the lead can act as a "team integrator" (Pianese et al., 2023, p. 337). Besides, formal procedures and the availability of ICTs can increase trust among team members by ensuring effective interactions and the pursuit of common goals (Olson, Appunn, McAllister, Walters, & Grinnell, 2014; Pianese et al., 2023; Thomas & Bostrom, 2008).

Another potential avenue for organizational restructuring to cultivate cognitive and relational social capital is to provide time and a shared physical space for global virtual teamwork (D. Cohen & Prusak, 2001). Setting that conduce personal interactions, such as cafés, chat rooms, libraries, or coffee kitchens, serve as ideal venues for sharing knowledge, exploring common interests, and establishing trust and mutual understanding (D. Cohen & Prusak, 2001; Prousak & Cohen, 2001). For managers overseeing global virtual teamwork, this implies a great need to create social spaces that facilitate the formation of cognitive and relational social capital when teams cannot conduct face-to-face meetings.

6.3. Research limitations and recommendations

Despite new theoretical and practical implications, it is essential to recognize the study's limitations. First, the chosen key informant approach bears the risk that a single representative cannot reflect the majority opinion of an entire global virtual team (Marshall, 1996). For this reason, it would be interesting to interview several team members, resulting in a different methodological approach in the form of case studies. Furthermore, relying on the subjective assessments of respondents introduces bias (Maier, Thatcher, Grover, & Dwivedi, 2023). Although several measures were taken to minimize biases, such as ensuring data confidentiality and mitigating socially desirable responses, future research could incorporate objective, secondary data on global virtual teamwork to complement the findings. Second, the study focused specifically on technology-sensitive and aware companies in the DACH region, which may limit the generalizability of the findings. While the research setting was appropriate for addressing the research question, and control variables were included in the research model to account for external influences on team effectiveness, it is necessary to conduct further research to explore potential variations in the findings across industries or countries. Third, the sample predominantly consisted of male respondents, reflecting the sex composition commonly found in manufacturing firms in developed countries (Cropley & Cropley, 2017; Reshef, Aneja, & Subramani, 2021). Although respondent sex was controlled for in the research model, future research could specifically investigate the role of sex in global virtual teamwork. Fourth, the study gathered cross-sectional data from a single respondent within a homogeneous industry and region, which may limit the generalizability of the findings. To enhance the transferability and robustness of the results across different industries and countries, future research could adopt a longitudinal approach and collect data from multiple respondents based on the research model proposed in this study.

7. Conclusion

The confluence of digitalization and globalization has had an enduring influence on utilizing virtual and global collaboration in the workplace. Due to their embeddedness in a social context, teams are a focal point of multiple levels, linking the individual team members and the organizational system holistically (Kozlowski & Bell, 2013). This requires a profound comprehension of the social context, which can enhance the uptake of ICTs and the effectiveness of global virtual teamwork. To fill this research gap, we conceptualized social capital as a multidimensional mediator, encompassing structural, cognitive, and relational social capital (Nahapiet & Ghoshal, 1998), between the adoption of ICTs and team effectiveness composed of performance, satisfaction, and longevity.

In summary, this study provides a solid foundation for integrating social capital in global virtual teamwork. The results of our study indicate that frequent use of multiple ICTs does not influence the effectiveness of global virtual teams. Instead, a significant positive relationship was found between ICT use and social capital. Regarding the individual dimensions of social capital, this relationship was only found for structural social capital. Similar results were obtained for the mediator hypotheses, with both total and structural social capital acting as strengthening mediators between ICT usage and team effectiveness. Conversely, cognitive and relational social capital did not show significant mediating effects. The present research context makes developing cognitive and relational social capital difficult, although their importance cannot be denied, as both social capital and its individual dimensions show a positive relationship with team effectiveness. Possible ways to build cognitive and relational social capital include allocating sufficient time for team interactions and creating shared physical spaces, such as face-to-face meetings, to balance them. Overall, our study provides insights into the extent to which the integration of social capital enhances team effectiveness in today's digitized and globalized world.

Declarations

Consent to participate. Consent was obtained from all individual participants included in the study.

Funding. This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

CRediT authorship contribution statement

Tim Heubeck: Writing – review & editing, Writing – original draft, Visualization, Supervision, Project administration, Methodology, Formal analysis, Data curation, Conceptualization. **Anne-Sophie Storz:** Writing – review & editing, Writing – original draft, Formal analysis, Data curation, Conceptualization. **Reinhard Meckl:** Supervision, Project administration.

Declaration of Generative AI and AI-assisted technologies in the writing process

During the preparation of this work, the authors used DeepL Write (https://www.deepl.com/write), ChatGPT (https://chat.openai.com/), and Grammarly (https://www.grammarly.com/) in order to provide suggestions for revising the formulations of this manuscript and check its grammatical correctness. After using these tools/services, the authors reviewed and edited the content as needed and took full responsibility for the publication's content.

Declaration of Competing Interest

The authors have no conflicts of interest to declare.

Data availability

Data will be made available on request.

Appendix

Table A1

Questionnaire items (English version).

Variable	Dimension	Item	
Team effectiveness Adapted from Wageman et al.	Performance	1 2	Members demonstrate their commitment to our team by putting in extra time and effort to help it succeed. Everyone on this team is motivated to have the team succeed.
(2000)		3	Some members of our team do not carry their fair share of the overall workload. (B)
		4	Our team often comes up with innovative ways of proceeding with the work that turns out to be just what is needed.
		5	Our team often falls into mindless routines without noticing any changes that may have occurred in our situation. (R)
		6	Our team has a great deal of difficulty actually carrying out the plans we make for how we will proceed with the task. (B)
		7	How seriously a member's ideas are taken by others on our team often depends more on who the person is than on
			how much he or she actually knows. (R)
		8	Members of our team actively share their special knowledge and expertise with one another.
		9	Our team is quite skilled at capturing the lessons that can be learned from our work experiences.
	Satisfaction	1	I feel a real sense of personal satisfaction when our team does well.
		2	I feel bad and unhappy when our team has performed poorly.
		3	My own feelings are not affected one way or the other by how well our team performs. (R)
		4	When our team has done well, I have done well.
		5	I learn a great deal from my work on this team.
		6	My own creativity and initiative are suppressed by this team. (R)
		7	Working on this team stretches my personal knowledge and skills.
		8	I enjoy the kind of work we do in this team.
		9	Working on this team is an exercise in frustration. (R)
		10	Generally speaking, I am very satisfied with this team.
	Longevity	1	There is a lot of unpleasantness among members of this team. (R)
		2	The longer we work together as a team, the less well we do. (R)
		3	Working together energizes and uplifts members of our team.
		4	Every time someone attempts to correct a team member whose behavior is not acceptable, things seem to get worse rather than better. (R)
		5	My relations with other team members are strained. (R)
		6	I very much enjoy talking and working with my teammates.
		7	The chance to get to know my teammates is one of the best parts of working on this team.
Social capital	Structural social capital	1	In our team, we maintain close social relationships with each other.
Adapted from Sun et al. (2012)	*	2	In our team, we spend a lot of time interacting with other team members.
()		3	In our team, we know some team members at a personal level.
		4	In our team, we have frequent communication with other team members.
	Cognitive social capital	1	When interacting in our team, we use common terms or jargon.
	-	2	During the discussion in our team, we use understandable communication patterns.
		3	When communicating in our team, we use understandable narrative forms.
	Relational social capital	1	The relationship is characterized by mutual respect in our team.
		2	The relationship is characterized by personal friendship in our team.
		3	The relationship is characterized by mutual trust in our team.
		4	The relationship is characterized by high reciprocity in our team.
ICT usage		Pleas	e indicate to what extent you have been using the following communication technologies to communicate with other bers of your team:
Adapted from Eisenberg et al. (2021)		1	Instant text-based messaging (chat, e.g., MSN, WhatsApp)
		2	Tele-conferencing (using voice and audio-based devices, such as telephone, etc.)
		3	Video-conferencing (voice and video-based devices, such as video-call on Microsoft Teams, etc.)
		4	Emails and social networks (e.g., Facebook)

Notes: (R) = reverse coded

Table B1

Results of confirmatory factor analysis.

Factor	Ν	KMO	Bartlett significance	AVE	FLR	α	Item	Std. FL
Structural social capital	314	0.733	< 0.001	0.470	0.363	0.774	1	0.714
-							2	0.783
							3	0.559
							4	0.667
Cognitive social capital	309	0.655	< 0.001	0.558	0.259	0.773	1	0.562
							2	0.884
							3	0.760
Relational social capital	311	0.665	< 0.001	0.480	0.692	0.710	1	0.735
							3	0.759
							4	0.568
Team performance (Factor 1)	302	0.792	< 0.001	0.415	0.911	0.776	1	0.536
							2	0.713
							4	0.627
							8	0.613
							9	0.715
Team performance (Factor 2)	303	0.638	< 0.001	0.415	0.684	0.666	5	0.641
							6	0.767
							7	0.502
Team satisfaction (Factor 1)	300	0.668	< 0.001	0.434	0.124	0.695	1	0.601
							2	0.697
							3	0.674
Team satisfaction (Factor 2)	299	0.500	< 0.001	0.608	0.383	0.757	5	0.780
							7	0.780
Team satisfaction (Factor 3)	298	0.728	< 0.001	0.467	0.829	0.763	6	0.600
							8	0.779
							9	0.495
							10	0.810
Team longevity (Factor 1)	294	0.787	< 0.001	0.509	0.707	0.805	1	0.764
							2	0.702
							4	0.685
							5	0.700
Team longevity (Factor 2)	295	0.633	< 0.001	0.503	0.752	0.726	3	0.655
							6	0.895
							7	0.529

Notes: α = Cronbach's alpha; AVE = Average variance extracted; FLR = Fornell-Larcker-ratio; KMO = Kaiser-Meyer-Olkin; *N* = Sample size; Std. FL = Standardized factor loading

Table C1

Regression results: Social capital model.

	DV: Social capital				DV: Team effectiveness						
Control variables			Study variables		Control variables		Study variables				
	b (se)	β	b (se)	β	b (se)	β	b (se)	β			
Study variables											
ICT usage			0.217** (0.071)	0.217**			0.061 (0.046)	0.060			
Social capital							0.707*** (0.048)	0.702***			
Control variables											
Firm size	-0.019 (0.017)	-0.070	-0.017 (0.017)	-0.065	-0.008 (0.017)	-0.031	0.005 (0.012)	0.020			
Team size	0.001 (0.001)	0.037	0.001 (0.001)	0.026	0.000 (0.001)	0.025	-0.000 (0.001)	-0.004			
Team age	-0.000 (0.000)	-0.004	-0.000 (0.000)	-0.036	0.000 (0.000)	0.028	0.000 (0.001)	0.022			
Team familiarity	0.159* (0.070)	0.143*	-0.150* (0.074)	-0.135*	0.081 (0.071)	0.072	-0.034 (0.054)	-0.030			
Team type	-0.047 (0.026)	-0.116	-0.056* (0.026)	-0.137*	-0.040 (0.026)	-0.097	-0.009 (0.020)	-0.021			
Respondent age	0.001 (0.003)	0.020	0.002 (0.003)	0.035	0.006 (0.003)	0.125	0.006* (0.003)	0.115*			
Respondent sex	0.014 (0.072)	0.012	0.027 (0.071)	0.028	0.012 (0.075)	0.011	0.008 (0.052)	0.007			
Respondent education	0.042 (0.040)	0.063	0.039 (0.043)	0.059	0.000 (0.041)	0.001	-0.030 (0.027)	-0.045			
Respondent nationality	-0.067 (0.065)	-0.063	-0.085 (0.081)	-0.080	-0.010 (0.066)	-0.010	0.032 (0.050)	0.030			
Respondent team tenure	0.001 (0.000)	0.121	0.001 (0.000)	0.118	-0.000 (0.000)	-0.016	-0.001 (0.000)	-0.102			
Constant	-0.024 (0.238)		-0.800* (0.340)		-0.213 (0.244)		-0.413 (0.241)				
R^2	0.063		0.107***		0.031		0.514***				
р	0.071		< 0.001		0.591		< 0.001				
Ν	271		271		271		271				

Notes: b = regression coefficient, $\beta =$ standardized regression coefficient, DV = dependent variable, N = sample size, p = significance value, $R^2 =$ coefficient of determination, se = standard error, ***p < 0.001, **p < 0.01, *p < 0.05; number of bootstrap samples = 5000, bootstrap inference for model coefficients, heteroscedasticity-consistent standard errors HC4 (Cribari-Neto)

Table D1

Regression results: Social capital dimensions model.

	DV: Structural social capital		DV: Cognitive soc	ial capital	DV: Relational so	cial capital	DV: Team effectiveness		
	b (se)	β	b (se)	β	b (se)	β	b (se)	β	
Study variables									
ICT usage	0.583*** (0.104)	0.360***	-0.019 (0.117)	-0.012	0.086 (0.091)	0.056	0.077 (0.046)	0.076	
Structural social capital							0.190*** (0.031)	0.306***	
Cognitive social capital							0.174*** (0.029)	0.286***	
Relational social capital							0.343*** (0.036)	0.523***	
Control variables									
Firm size	-0.032 (0.025)	-0.075	0.001 (0.028)	0.002	-0.021 (0.027)	-0.050	0.006 (0.012)	0.022	
Team size	0.004* (0.002)	0.110*	-0.002 (0.003)	-0.061	0.000 (0.002)	0.001	0.000 (0.002)	-0.003	
Team age	0.000 (0.001)	0.016	-0.000 (0.001)	-0.045	-0.000 (0.001)	-0.038	0.000 (0.001)	0.025	
Team familiarity	-0.377** (0.114)	-0.210**	0.073 (0.122)	0.010	-0.000 (0.114)	-0.000	-0.012 (0.052)	-0.011	
Team type	-0.071 (0.038)	-0.107	-0.037 (0.046)	-0.054	-0.060 (0.041)	-0.096	-0.008 (0.019)	-0.019	
Respondent age	-0.008 (0.005)	-0.100	0.007 (0.006)	0.084	0.007 (0.006)	0.085	0.005 (0.003)	0.102	
Respondent sex	0.095 (0.107)	0.051	0.091 (0.122)	0.048	-0.090 (0.121)	-0.051	0.027 (0.049)	0.023	
Respondent education	0.034 (0.063)	0.032	0.062 (0.075)	0.057	0.021 (0.067)	0.021	-0.027 (0.028)	-0.040	
Respondent nationality	-0.193 (0.130)	-0.112	-0.005 (0.122)	-0.003	-0.056 (0.128)	-0.034	0.029 (0.047)	0.027	
Respondent team tenure	0.001 (0.001)	0.089	0.000 (0.001)	0.044	0.001 (0.001)	0.088	-0.001 (0.000)	-0.106	
Constant	-1.670*** (0.519)		-0.364 (0.552)		-0.366 (0.549)		-0.472* (0.240)		
R^2	0.231***		0.022		0.038		0.549***		
р	< 0.001		0.956		0.376		< 0.001		
Ν	271		271		271		271		

Notes: b = regression coefficient, $\beta =$ standardized regression coefficient, DV = dependent variable, N = sample size, p = significance value, $R^2 =$ coefficient of determination, se = standard error, **p < 0.001, *p < 0.05; number of bootstrap samples = 5000, bootstrap inference for model coefficients, heteroscedasticity-consistent standard errors HC4 (Cribari-Neto)

Table E1

Bootstrapping regression results and Sobel's test.

Total effect	Indirect effects	b (se)	Confidence interval			β (se)	Confidence interval			Sobel's test	
			Lower	Upper	Level		Lower	Upper	Level	b	р
ICT usage → Team effectiveness		0.214** (0.069)	0.035	0.393	99%						
	ICT usage → Social capital → Team effectiveness	0.153** (0.049)	0.023	0.290	99%	0.152** (0.047)	0.031	0.277	99%	0.153**	0.003
	ICT usage \rightarrow Structural social capital \rightarrow Team effectiveness	0.111** (0.026)	0.052	0.189	99%	0.110** (0.025)	0.055	0.181	99%	0.111***	< 0.001
	ICT usage \rightarrow Cognitive social capital \rightarrow Team effectiveness	-0.003 (0.021)	-0.037	0.030	90%	-0.003 (0.021)	-0.037	0.029	90%	-0.003	0.872
	ICT usage \rightarrow Relational social capital \rightarrow Team effectiveness	0.029 (0.031)	-0.022	0.079	90%	0.029 (0.030)	-0.021	0.078	90%	0.029	0.349

Notes: se = Standard error, b = coefficient, N = Sample size; ***p < 0.001, **p < 0.01, *p < 0.05; number of bootstrap samples = 5000, bootstrap inference for model coefficients, heteroscedasticity-consistent standard errors HC4 (Cribari-Neto), N = 271.

References

- Adler, P. S., & Kwon, S.-W. (2002). Social capital: Prospects for a new concept. Academy of Management Review, 27(1), 17–40. https://doi.org/10.5465/amr.2002.5922314Adner, R., & Helfat, C. E. (2003). Corporate effects and dynamic managerial capabilities.
- Strategic Management Journal, 24, 1011–1025. https://doi.org/10.1002/smj.331 Ahuja, G. (2000). Collaboration networks, structural holes, and innovation: A
- Iongitudinal study. Administrative Science Quarterly, 45(3), 425–455. https://doi.org/ 10.2307/2667105
- Ali-Hassan, H., Nevo, D., & Wade, M. (2015). Linking dimensions of social media use to job performance: The role of social capital. *Journal of Strategic Information Systems*, 24(2), 65–89. https://doi.org/10.1016/j.jsis.2015.03.001
- Andersson, U., Forsgren, M., & Holm, U. (2002). The strategic impact of external networks: Subsidiary performance and competence development in the multinational corporation. *Strategic Management Journal*, 23, 979–996. https://doi. org/10.1002/smj.267
- Badir, Y. F., Büchel, B., & Tucci, C. L. (2012). A conceptual framework of the impact of NPD project team and leader empowerment on communication and performance: An alliance case context. *International Journal of Project Management, 30*, 914–926. https://doi.org/10.1016/j.ijproman.2012.01.013
- Badir, Y. F., & O'Connor, G. C. (2015). The formation of tie strength in a strategic Alliance's first new product development project: The influence of project and Partners' characteristics. *Journal of Product Innovation Management*, 32(1), 154–169. https://doi.org/10.1111/jpim.12222

- Badrinarayanan, V., Madhavaram, S., & Granot, E. (2011). Global virtual sales teams (GVSTs): A conceptual framework of the influence of intellectual and social capital on effectiveness. *Journal of Personal Selling & Sales Management*, 31(3), 311–324. https://doi.org/10.2753/PSS0885-3134310308
- Baron, R. M., & Kenny, D. A. (1986). The moderator-mediator variable distinction in social psychological research: Conceptual, strategic, and statistical considerations. *Journal of Personality and Social Psychology*, 51(6), 1173–1182. https://doi.org/ 10.1037//0022-3514.51.6.1173
- Beaton, D., Bombardier, C., Guillemin, F., & Ferraz, M. (2007). Recommendations for the cross-cultural adaptation of the DASH & QuickDASH outcome measures. https:// www.semanticscholar.org/paper/Recommendations-for-the-Cross-Cultural-Adaptat ion-%26-Beaton-Bombardier/0f67d948651f607e865ed66aafc23805c52a3272.
- Bell, B. S., & Kozlowski, S. W. J. (2002). A typology of virtual teams: Implications for effective leadership. Group & Organization Management, 27(1), 14–49. https://doi. org/10.1177/1059601102027001003
- Bilotta, I., Cheng, S. K., Ng, L. C., Corrington, A. R., Watson, I., Paoletti, J., ... King, E. B. (2021). Remote communication amid the coronavirus pandemic: Optimizing interpersonal dynamics and team performance. *Industrial and Organizational Psychology*, 14(1–2), 36–40, https://doi.org/10.1017/jop.2021.10
- Psychology, 14(1–2), 36–40. https://doi.org/10.1017/iop.2021.10
 Boland, R. J., & Tenkasi, R. V. (1995). Perspective making and perspective taking in communities of knowing. Organization Science, 6(4), 350–372. http://www.jstor.org/stable/2634993.
- Bourdieu, P. (1986). The forms of capital. In J. G. Richardson (Ed.), Handbook of theory and research for the sociology of education (1st ed., pp. 241–258). Greenwood Press.

T. Heubeck et al.

Bstieler, L. (2006). Trust formation in collaborative new product development. Journal of Product Innovation Management, 23, 56–72. https://doi.org/10.1111/j.1540-5885.2005.00181.x

- Cao, X., Guo, X., Vogel, D., & Zhang, X. (2016). Exploring the influence of social media on employee work performance. *Internet Research*, 26(2), 529–545. https://doi.org/ 10.1108/IntR-11-2014-0299
- Cao, X., Vogel, D. R., Guo, X., Liu, H., & Gu, J. (2012). Understanding the influence of social Media in the Workplace: An integration of media synchronicity and social capital theories. In 45th Hawaii international conference on system sciences (pp. 3938–3947). https://doi.org/10.1109/HICSS.2012.618
- Castro, I., & Roldán, J. L. (2013). A mediation model between dimensions of social capital. International Business Review, 22(6), 1034–1050. https://doi.org/10.1016/j. ibusrev.2013.02.004
- Chan, M. (2015). Mobile phones and the good life: Examining the relationships among mobile use, social capital and subjective well-being. *New Media & Society*, 17(1), 96–113. https://doi.org/10.1177/1461444813516836
- Chang, H. H., & Chuang, S.-S. (2011). Social capital and individual motivations on knowledge sharing: Participant involvement as a moderator. *Information & Management*, 48(1), 9–18. https://doi.org/10.1016/j.im.2010.11.001
- Chen, H.-T., & Li, X. (2017). The contribution of mobile social media to social capital and psychological well-being: Examining the role of communicative use, friending and self-disclosure. *Computers in Human Behavior*, 75, 958–965. https://doi.org/ 10.1016/j.chb.2017.06.011
- Cohen, D., & Prusak, L. (2001). In Good Company: In good company: How Social Capital Makes Organizations Work. Harvard Business School Press.
- Cohen, J. (1988). Statistical power analysis for the behavioral sciences (2nd ed.). L. Erlbaum Associates.
- Cohen, S. G., & Bailey, D. E. (1997). What makes teams work: Group effectiveness research from the shop floor to the executive suite. *Journal of Management*, 23(3), 239–290. https://doi.org/10.1177/014920639702300303
- Coleman, J. S. (1988). Social Capital in the Creation of human capital. American Journal of Sociology, 94, S95–S120. http://www.jstor.org/stable/2780243.
- Comrey, A. L., & Lee, H. B. (1992). A first course in factor analysis (2nd ed.). Psychology Press.
- Cropley, D., & Cropley, A. (2017). Innovation capacity, organisational culture and gender. European Journal of Innovation Management, 20(3), 493–510. https://doi. org/10.1108/EJIM-12-2016-0120
- Daft, R. L., & Lengel, R. H. (1986). Organizational information requirements, media richness and structural design. *Management Science*, 32(5), 554–571. http://www. jstor.org/stable/2631846.
- Dennis, A. R., Fuller, R. M., & Valacich, J. S. (2008). Media, tasks, and communication processes: A theory of media synchronicity. *MIS Quarterly*, 32(3), 575–600. https:// doi.org/10.2307/25148857
- Dennis, A. R., Valacich, J., Speier, C., & Morris, M. G. (1998). Beyond media richness: An empirical test of media synchronicity theory. In *Proceedings 31st annual Hawaii* international conference on system sciences. https://doi.org/10.1109/ HICSS.1998.653082. Advance online publication.
- van Dijk, B. (2023). Amadeus | European business information | bureau van Dijk. https://www.bvdinfo.com/en-gb/our-products/data/international/amadeus. Earley, P. C., & Mosakowski, E. (2000). Creating hybrid team cultures: An empirical test
- of transnational team functioning. Academy of Management Journal, 43(1), 26-49. Eisenberg, J., Glikson, E., & Lisak, A. (2021). Multicultural virtual team performance:
- The impact of media choice and language diversity. *Small Group Research*, 52(5), 507–534. https://doi.org/10.1177/1046496420985614
- Federal Anti-Discrimination Agency. (2018). Male female diverse: The "third option" and the general act on equal treatment. Federal Anti-Discrimination Agency. https:// www.antidiskriminierungsstelle.de/EN/about-discrimination/grounds-for-discri mination/gender-and-gender-identity/third-option/third-option-node.html.
- Flap, H., & Völker, B. (2001). Goal specific social capital and job satisfaction. Social Networks, 23(4), 297–320. https://doi.org/10.1016/S0378-8733(01)00044-2
- Fornell, C., & Larcker, D. F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *Journal of Marketing Research*, 18(1), 39. https://doi.org/10.2307/3151312
- Fornoni, M., Arribas, I., & Vila, J. E. (2011). Measurement of an individual entrepreneur's social capital: A multidimensional model. *International Entrepreneurship and Management Journal*, 7(4), 495–507. https://doi.org/10.1007/ s11365-011-0204-1
- Fu, J., Sawang, S., & Sun, Y. (2019). Enterprise social media adoption: Its impact on social Capital in Work and job Satisfaction. *Sustainability*, 11(16), 1–17. https://doi. org/10.3390/su11164453
- Gargiulo, M., & Benassi, M. (1999). The dark side of social capital. In R. T. A. J. Leenders, & S. M. Gabbay (Eds.), *Corporate social capital and liability* (pp. 298–322). Kluwer Academic Publishers.
- Gedajlovic, E., Honig, B., Moore, C. B., Payne, G. T., & Wright, M. (2013). Social capital and entrepreneurship: A Schema and research agenda. *Entrepreneurship Theory and Practice*, 37(3), 455–478. https://doi.org/10.1111/etap.12042
- Gefen, K., & Straub. (2003). Trust and TAM in online shopping: An integrated model. *MIS Quarterly*, *27*(1), 51–90. https://doi.org/10.2307/30036519
- Ghoshal, S., & Moran, P. (1996). Bad for practice: A critique of the transaction cost theory. Academy of Management Review, 21(1), 13–47. https://doi.org/10.2307/ 258627
- Gosling, S. D., Rentfrow, P. J., & Swann, W. B. (2003). A very brief measure of the bigfive personality domains. *Journal of Research in Personality*, 37(6), 504–528. https:// doi.org/10.1016/S0092-6566(03)00046-1

- Gower, J. C. (1966). Some distance properties of latent root and vector methods used in multivariate analysis. *Biometrika*, 53(3/4), 325–338. https://doi.org/10.2307/ 2333639
- Grooss, O. F., Presser, M., & Tambo, T. (2022). Surround yourself with your betters: Recommendations for adopting industry 4.0 technologies in SMEs. *Digital Business, 2* (2), Article 100046. https://doi.org/10.1016/j.digbus.2022.100046
- Guillemin, F., Bombardier, C., & Beaton, D. (1993). Cross-cultural adaptation of healthrelated quality of life measures: Literature review and proposed guidelines. *Journal* of *Clinical Epidemiology*, 46(12), 1417–1432. https://doi.org/10.1016/0895-4356 (93)90142-n
- Gupta, S., Modgil, S., Gunasekaran, A., & Bag, S. (2020). Dynamic capabilities and institutional theories for industry 4.0 and digital supply chain. *Supply chain forum: An. International Journal*, 21(3), 139–157. https://doi.org/10.1080/ 16258312.2020.1757369
- Gupta, V., & Hanges, P. J. (2006). Regional and climate clustering of societal cultures. In R. J. House, P. J. Hanges, M. Javidan, P. W. Dorfman, & V. Gupta (Eds.), *Culture, leadership, and organizations: The GLOBE study of 62 societies* (pp. 178–218). Sage Publications.
- Hackman, J. R. (1987). The design of work teams. In J. W. Lorsch (Ed.), Handbook of organizational behavior (pp. 315–342). Prentice-Hall.
- Hair, J. F., Black, W. C., Babin, B. J., & Anderson, R. E. (2014). Multivariate data analysis (7th ed.). Pearson.
- Harkness, J., Pennell, B.-E., & Schoua-Glusberg, A. (2004). Survey Questionnaire Translation and Assessment. In S. Presser, J. M. Rothgeb, M. P. Couper, J. T. Lessler, E. Martin, J. Martin, & E. Singer (Eds.), Wiley series in survey methodology. Methods for testing and evaluating survey questionnaires (pp. 453–473). Wiley-Interscience. https://doi.org/10.1002/0471654728.ch22.

Harman, H. H. (1976). Modern factor analysis (3rd ed.). Univ. of Chicago Press.

- Hayes, A. F. (2009). Beyond Baron and Kenny: Statistical mediation analysis in the new millennium. Communication Monographs, 76(4), 408–420. https://doi.org/10.1080/ 03637750903310360
- Heubeck, T. (2023). Managerial capabilities as facilitators of digital transformation? Dynamic managerial capabilities as antecedents to digital business model transformation and firm performance. *Digital Business*, 3(1), Article 100053. https:// doi.org/10.1016/j.digbus.2023.100053
- Heubeck, T., & Meckl, R. (2022). More capable, more innovative? An empirical inquiry into the effects of dynamic managerial capabilities on digital firms' innovativeness. *European Journal of Innovation Management*, 25(6), 892–915. https://doi.org/ 10.1108/EJIM-02-2022-0099
- Hofstede, G. (2005). Culture's consequences: Comparing values, behaviors, institutions, and organizations across nations (2nd ed.). Sage Publications.
- Huang, Q., Chen, X., Ou, C. X., Davison, R. M., & Hua, Z. (2017). Understanding buyers' loyalty to a C2C platform: The roles of social capital, satisfaction and perceived effectiveness of e-commerce institutional mechanisms. *Information Systems Journal*, 27, 91–119. https://doi.org/10.1111/isj.12079
- Huysman, M., & Wulf, V. (2006). IT to support knowledge sharing in communities, towards a social capital analysis. *Journal of Information Technology*, 21, 40–51. https://doi.org/10.1057/palgrave.jit.2000053
- Jarvenpag, S. L., & Leidner, D. E. (1999). Communication and Trust in Global Virtual Teams. Organization Science, 10(6), 792–815.
- Jarvenpaa, S. L., Shaw, T. R., & Staples, D. S. (2004). Toward contextualized theories of trust: The role of Trust in Global Virtual Teams. *Information Systems Research*, 15(3), 250–267. https://doi.org/10.1287/isre.1040.0028
- Jimenez, A., Boehe, D. M., Taras, V., & Caprar, D. V. (2017). Working across boundaries: Current and future perspectives on global virtual teams. *Journal of International Management*, 23(4), 341–349. https://doi.org/10.1016/j.intman.2017.05.001
- Johnston, R., Jones, K., & Manley, D. (2018). Confounding and collinearity in regression analysis: A cautionary tale and an alternative procedure, illustrated by studies of British voting behaviour. *Quality & Quantity*, 52(4), 1957–1976. https://doi.org/ 10.1007/s11135-017-0584-6
- Jones, T., & Taylor, S. F. (2012). Service loyalty: Accounting for social capital. Journal of Services Marketing, 26(1), 60–74. https://doi.org/10.1108/08876041211199733
- Kamboj, S., Kumar, V., & Rahman, Z. (2017). Social media usage and firm performance: The mediating role of social capital. *Social Network Analysis and Mining*, 7(1), 1–14. https://doi.org/10.1007/s13278-017-0468-8
- Kirkman, B. L., Cordery, J. L., Mathieu, J., Rosen, B., & Kukenberger, M. (2013). Global organizational communities of practice: The effects of nationality diversity, psychological safety, and media richness on community performance. *Human Relations*, 66(3), 333–362. https://doi.org/10.1177/0018726712464076
- Klitmøller, A., & Lauring, J. (2013). When global virtual teams share knowledge: Media richness, cultural difference and language commonality. *Journal of World Business*, 48(3), 398–406. https://doi.org/10.1016/j.jwb.2012.07.023
- Knight, M. B., Pearson, J. M., & Hunsinger, D. S. (2008). The role of media richness in information technology-supported communication in group cohesion, agreeability, and performance. *Journal of Organizational and End User Computing*, 20(4), 23–44. https://doi.org/10.4018/joeuc.2008100102
- Kock, N. (1998). Can communication medium limitations foster better group outcomes? An action research study. *Information & Management*, 34(5), 295–305. https://doi. org/10.1016/S0378-7206(98)00066-4
- Kozlowski, S. W. J., & Bell, B. S. (2013). Chapter 17 chapter 17: Work groups and teams in organizations. In I. B. Weiner (Ed.), *Handbook of psychology: Volume 12: Industrial* and organizational psychology (2nd ed., pp. 412–469). John Wiley & Sons, Inc.
- Kozlowski, S. W. J., & Ilgen, D. R. (2006). Enhancing the effectiveness of work groups and teams. Psychological Science in the Public Interest : A Journal of the American Psychological Society, 7(3), 77–124. https://doi.org/10.1111/j.1529-1006.2006.00030.x

- Lee, J. Y.-H., Saunders, C., Panteli, N., & Wang, T. (2021). Managing information sharing: Interorganizational communication in collaborations with competitors. *Information and Organization*, 31(2), Article 100354. https://doi.org/10.1016/j. infoandorg.2021.100354
- Lesser, E. L., & Storck, J. (2001). Communities of practice and organizational performance. *IBM Systems Journal*, 40(4), 831–841. https://doi.org/10.1147/ sj.404.0831
- Lin, N. (2001). Social capital: A theory of social structure and action. Structural analysis in the social sciences (Vol. 19). Cambridge university press.
- Liu, C.-L. (2012). An investigation of relationship learning in cross-border buyer-supplier relationships: The role of trust. *International Business Review*, 21(3), 311–327. https://doi.org/10.1016/j.ibusrev.2011.05.005
- MacKenzie, S. B., & Podsakoff, P. M. (2012). Common method Bias in marketing: Causes, mechanisms, and procedural remedies. *Journal of Retailing*, 88(4), 542–555. https:// doi.org/10.1016/j.jretai.2012.08.001
- Maier, C., Thatcher, J. B., Grover, V., & Dwivedi, Y. K. (2023). Cross-sectional research: A critical perspective, use cases, and recommendations for IS research. *International Journal of Information Management*, 70, 102625. https://doi.org/10.1016/j. ijinfomgt.2023.102625
- Mangla, N. (2021). Working in a pandemic and post-pandemic period Cultural intelligence is the key. *International Journal of Cross Cultural Management*, 21(1), 53–69. https://doi.org/10.1177/14705958211002877
- Marshall, M. N. (1996). The key informant technique. Family Practice, 13(1), 92–97. https://doi.org/10.1093/fampra/13.1.92
- McLure Wasko, M., & Faraj, S. (2005). Why should I share? Examining social capital and knowledge contribution in electronic networks of practice. *MIS Quarterly*, 29(1), 35–37. https://doi.org/10.2307/25148667
- Menard, S. (2002). Applied logistic regression analysis. SAGE Publications, Inc.. https://me thods.sagepub.com/book/applied-logistic-regression-analysis. https://doi.org/10.41 35/9781412983433
- Nahapiet, J., & Ghoshal, S. (1998). Social capital, intellectual capital, and the organizational advantage. Academy of Management Review, 23(2), 242–266. https:// doi.org/10.2307/259373
- Niinimäki, T., Piri, A., Lassenius, C., & Paasivaara, M. (2012). Reflecting the choice and usage of communication tools in global software development projects with media synchronicity theory. *Journal of Software: Evolution and Process*, 24(6), 677–692. https://doi.org/10.1002/smr.566
- Olson, J. D., Appunn, F. D., McAllister, C. A., Walters, K. K., & Grinnell, L. (2014). Webcams and virtual teams: An impact model. *Team Performance Management*, 20(3/ 4), 148–177. https://doi.org/10.1108/TPM-05-2013-0013
- Osborne, J. W., & Costello, A. B. (2019). Sample size and subject to item ratio in principal components analysis. *Practical Assessment, Research and Evaluation*, 9(1). https://doi. org/10.7275/ktzq-jq66
- Pang, H. (2018). Understanding the effects of WeChat on perceived social capital and psychological well-being among Chinese international college students in Germany. *Aslib Journal of Information Management*, 70(3), 288–304. https://doi.org/10.1108/ AJIM-01-2018-0003
- Park, S. H., & Luo, Y. (2001). Guanxi and organizational dynamics: Organizational networking in Chinese firms. *Strategic Management Journal*, 22, 455–477. https://doi. org/10.1002/smj.167
- Parlamis, J., & Dibble, R. (2019). Teaming: Are two communication modes better than one? Team Performance Management: An International Journal, 25(5/6), 318–333. https://doi.org/10.1108/TPM-10-2018-0065
- Paulraj, A., Lado, A. A., & Chen, I. J. (2008). Inter-organizational communication as a relational competency: Antecedents and performance outcomes in collaborative buyer-supplier relationships. *Journal of Operations Management, 26*, 45–64. https:// doi.org/10.1016/j.jom.2007.04.001
- Pfister, J. (2009). Managing organizational culture for effective internal control: From Practic to theory. Physica-Verlag. https://doi.org/10.1007/978-3-7908-2340-0
 Pianese, T., Errichiello, L., & Da Cunha, J. V. (2023). Organizational control in the
- Pianese, T., Errichiello, L., & Da Cunha, J. V. (2023). Organizational control in the context of remote working: A synthesis of empirical findings and a research agenda. *European Management Review*, 20(2), 326–345. https://doi.org/10.1111/emre.12515
- Podsakoff, P. M., MacKenzie, S. B., Lee, J.-Y., & Podsakoff, N. P. (2003). Common method biases in behavioral research: A critical review of the literature and recommended remedies. *The Journal of Applied Psychology*, 88(5), 879–903. https://doi.org/ 10.1037/0021-9010.88.5.879
- Price, J. L. (1997). Handbook of organizational measurement. International Journal of Manpower, 18(4/5/6), 305–558. https://doi.org/10.1108/01437729710182260
- Prousak, L., & Cohen, D. (2001). How to invest in social capital. *Harvard Business Review*, 79(6), 86–93.
- Putnam, R. D., Leonardi, R., & Nanetti, R. (1994). Making democracy work: Civic traditions in modern Italy. In *Princeton paperbacks: Political science* (5th ed.). Princeton University Press.
- Raghupathi, V., & Benbunan-Fich, R. (2020). A social capital perspective on computermediated group communication and performance: An empirical study. *Group Decision and Negotiation*, 29(4), 747–801. https://doi.org/10.1007/s10726-020-09678-y

- Reshef, O., Aneja, A., & Subramani, G. (2021). Persistence and the Gender Innovation Gap: Evidence from the U.S. Patent and Trademark Office. Academy of Management Proceedings, 2021(1), 11626. https://doi.org/10.5465/AMBPP.2021.11626abstract
- Riemer, K., & Klein, S. (2008). Is the V-form the next generation organisation? An analysis of challenges, pitfalls and remedies of ICT-enabled virtual organisations based on social capital theory. *Journal of Information Technology*, 23(3), 147–162. https://doi.org/10.1057/palgrave.jit.2000120
- Rowley, T., Behrens, D., & Krackhardt, D. (2000). Redundant governance structures: An analysis of structural and relational embeddedness in the steel and semiconductor industries. *Strategic Management Journal*, 21(3), 369–386. https://doi.org/10.1002/ (SICI)1097-0266(200003)21:3<369::AID-SMJ93>3.0.CO;2-M
- Rudolph, C. W., Allan, B., Clark, M., Hertel, G., Hirschi, A., Kunze, F., ... Zacher, H. (2021). Pandemics: Implications for research and practice in industrial and organizational psychology. *Industrial and Organizational Psychology*, 14, 1–35. https://doi.org/10.1017/iop.2020.48
- Schneider, P. (2018). Managerial challenges of industry 4.0: An empirically backed research agenda for a nascent field. *Review of Managerial Science*, 12(3), 803–848. https://doi.org/10.1007/s11846-018-0283-2
- Sobel, M. E. (1982). Asymptotic confidence intervals for indirect effects in structural equation models. Sociological Methodology, 13, 290. https://doi.org/10.2307/ 270723
- Staber, U. (2006). Social capital processes in cross cultural management. International Journal of Cross Cultural Management, 6(2), 189–203. https://doi.org/10.1177/ 1470595806066328
- Straube, J., Meinecke, A. L., Schneider, K., & Kauffeld, S. (2018). Effects of media compensation on team performance: The role of demographic Faultlines. *Small Group Research*, 49(6), 684–722. https://doi.org/10.1177/1046496418796281
- Sun, Y., Fang, Y., Lim, K. H., & Straub, D. (2012). User satisfaction with information technology service delivery: A social capital perspective. *Information Systems Research*, 23(4), 1195–1211. https://doi.org/10.1287/isre.1120.0421
- Tavoletti, E., & Taras, V. (2023). From the periphery to the centred: A bibliometric review of global virtual teams as a new ordinary workplace. *Management Research Review*, 46(8), 1061–1090.
- Thomas, D. M., & Bostrom, R. P. (2008). Building trust and cooperation through technology adaptation in virtual teams: Empirical field evidence. In Proceedings 41 St annual Hawaii international conference on system sciences. https://doi.org/10.1080/ 10580530701777149. Advance online publication.
- Thompson, B. (2004). Exploratory and confirmatory factor analysis: Understanding concepts and applications. *American Psychological Association*. https://doi.org/ 10.1037/10694-000
- Tsai, W., & Ghoshal, S. (1998). Social capital and value creation: The role of intrafirm networks. Academy of Management Journal, 41(4), 464–476. https://doi.org/ 10.2307/257085
- Tsang, S., Royse, C. F., & Terkawi, A. S. (2017). Guidelines for developing, translating, and validating a questionnaire in perioperative and pain medicine. *Saudi Journal of Anaesthesia*, 11(Suppl. 1), S80–S89. https://doi.org/10.4103/sja.SJA_203_17
- Uzzi, B. (1996). The sources and consequences of embeddedness for the economic performance of organizations: The network effect. *American Sociological Review*, 61 (4), 674–698. https://doi.org/10.2307/2096399
- Villena, V. H., Revilla, E., & Choi, T. Y. (2011). The dark side of buyer-supplier relationships: A social capital perspective. *Journal of Operations Management*, 29, 561–576. https://doi.org/10.1016/j.jom.2010.09.001
- Wageman, R., Hackman, J. R., & Lehman, E. (2005). Team diagnostic survey: Development of an instrument. *The Journal of Applied Behavioral Science*, 41(4), 373–398. https://doi.org/10.1177/0021886305281984
- Walther, J. B., & Parks, M. R. (2002). Cues filtered out, cues filtered in: computermediated communication and relationships. In M. L. Knapp, & J. A. Daly (Eds.), *Handbook of interpersonal communication* (3rd ed., pp. 529–563). Sage.
- Watson-Manheim, M. B., & Bélanger, F. (2007). Communication Media Repertoires: Dealing with the Multiplicity of Media Choices. *MIS Quarterly*, 31(2), 267–293. https://doi.org/10.2307/25148791
- Wooldridge, J. M. (2019). Introductory econometrics: A modern approach (7th ed.). Cengage.
- Wu, M., Coleman, M., Abdul Rahaman, A. R., & Edziah, B. K. (2020). Successor selection in family business using theory of planned behaviour and cognitive dimension of social capital theory: Evidence from Ghana. *Journal of Small Business and Enterprise Development*, 27(6), 905–926. https://doi.org/10.1108/JSBED-05-2019-0152
- Wu, W. (2008). Dimensions of social capital and firm competitiveness improvement: The mediating role of information sharing. *Journal of Management Studies*, 45(1), 122–145. https://doi.org/10.1111/j.1467-6486.2007.00741.x
- Yamaguchi, I. (2013). A Japan–US cross-cultural study of relationships among team autonomy, organizational social capital, job satisfaction, and organizational commitment. *International Journal of Intercultural Relations*, 37(1), 58–71. https:// doi.org/10.1016/j.ijintrel.2012.04.016
- Zhao, X., Lynch, J. G., & Chen, Q. (2010). Reconsidering Baron and Kenny: Myths and truths about mediation analysis. *Journal of Consumer Research*, 37(2), 197–206. https://doi.org/10.1086/651257